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	CRUISE: EX10-04 Leg II
Bitung to Bitung
	PERSONNEL ON BOARD:

NOAA OER:

1. Mashkoor Malik (ERT Inc)

2. Dave Lovalvo (Eastern Oceanics)

3. Webb Pinner (2020 LLC)

4. Jeremy Potter (OER)

Others:

5. Doug Jongeward (PMEL)

6. John Sherrin (UVic)

7. Xerandy (BPPT)

8. Major Muddan (TNI)

9. Dave Wright (UCAR/JOSS)

10. Joel DeMello (UCAR/JOSS)

11. Tom Kok (UCAR/JOSS)

12. Tom Pierce (UCAR/JOSS)

13. Karl McLetchie (UCAR/JOSS)

14. Brian Bingham (UCAR/JOSS)
15. Randy Prickett (UCAR/JOSS)

16. Eric Prechtl (UCAR/JOSS)

17. Roland Brian (UCAR/JOSS)

18. Joe Biscotti (UCAR/JOSS)

	DATE/TIME FILED:  07/18/2010 1122 UTC

FILED BY: Jeremy Potter
	

	VESSEL:  NOAA Ship Okeanos Explorer (EX)


	

	GEOGRAPHIC AREA:  In the Port of Bitung, Indonesia 
	


SUMMARY:    Since the previous SITREP on July 8th, we conducted ROV dives everyday between July 9 and July 13.  Multibeam operations took place at all other times. The video team continues to adapt imagery processing and is implementing a workaround to network difficulties to enable ship-to-shore transfer of ROV video and still imagery in a more timely fashion. Upon completion of the July 13 ROV dive, Major Muddan confirmed that all applicable shipboard scientific data collection sensors were turned off prior to departing the approved area of operations. The ship arrived in the Port of Bitung at approximately 0800 on Wednesday, July 14. 
NOTE: As of today, local ship time is 8 hrs ahead of UTC.  Jakarta is 7 hrs ahead of UTC. East Coast is 4 hrs behind UTC. West Coast is 7 hrs behind UTC.

============================================================================
ROV:
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July 9: The crew launched the ROV for dive 9 at 4º 56.96'N 125 º 46.70'E around 0815. The target was an oddly shaped feature that rose to a depth of approximately 500m and had an almost square shaped and flat plateau. The plateau measured roughly 4000 by 3500 meters. It dropped off steeply on all sides to a depth of 1100m below. The top surface was composed primarily of eroded carbonate rock with fine-grained pelagic sediment on top. We ascended up a steep slope that contained a high biomass of corals, crinoids, shrimp and galatheid crabs (i.e. squat lobsters). Two of the galatheid crabs were holding a cylindrical shaped purple object that appeared to be some type of egg case or plankton which we could not immediately identify. We saw many of these objects float by the camera throughout the dive. The top of the plateau was fairly deserted in terms of biomass. We did however come across two animals that appeared to be either octopi or squid.  We also found an approximately 1m long stingray and several urchins that appeared to have the ability to climb up corals and sponge stalks. After canvassing the plateau for a few hours we descended back down the slope in a different location. The slope included a number of interesting geological features and high biomass. We also found a school of fish and a sandy area beneath a cliff.

July 10: Before launching the ROV on initial planned location, the bridge discovered that the dive area was located in an explosive ordinance zone. The dive location was moved to a ridge to the south that rose from the bottom to a depth of approximately 2240m.  The crew launch the ROV for Dive 10 at 3º 16.0926'N 125º 17.454’E around 0845. There as almost no current at the bottom and the surface was heavily sedimented with very few animals anywhere. We did however come across a few very interesting creatures as we climbed up the slope. These included a large purple spoon worm whose digestive tract was clearly visible through its translucent skin; a fascinating large purple nudibranch; and a camouflaged sea urchin covered with a hat of organic debris and smaller animals. We also noticed a number of large burrows in the sediment and several track lines with a lot of oddly shaped debris or perhaps unidentified organism. Near the end of the dive we came across a small cliff face that resembled a canyon with pale orange colored rock, a stark contrast from the brown sediment we saw for most of the dive. NOTE: Keith Luse, Senator Lugar’s Senior Professional Staff Member for East Asian and Pacific Affairs at the U.S. Senate Foreign Relations Committee, observed ROV operations from the Jakarta ECC on July 10.

July 11: 
The crew launch the ROV for Dive 11 at 2º 51.76'N 125º 02.06’E around 0815.  The ROV descended to a volcanic cone shaped seamount hoping to find new vents and their associated critters. Instead we were astonished by the diversity of corals, sponges, crinoids, and countless fascinating invertebrates. Personnel were immediately struck by the incredible diversity of colors and species on display as we rose up along the steep WSW slope toward the summit. Some of the organisms imaged included a 30cm wide, 20cm tall sea spider (pycnogonid) whose bright orange spindly legs and specially modified arms were quite a sight to see; a colorful octopus who danced around frantically in search of a place to hide; a large flat fish with big protruding eyes; a striped crinoid; and a wide array of fish of different types and colors including one species that had little feeler fins beneath its pectoral fins. We also saw some large white venus flytrap anemones, many colorful corals of many different species and morphologies, a huge brittle star filter feeding on a rock, large tubular sponges with striped fish feeding inside them, and fluorescent blue mucous nets that stretched up to a meter long from rock to rock. Overall this site contained the greatest diversity of any site we have found thus.
July 12:
The crew launch the ROV for Dive 12 at 2º 37.480'N 125º 05.18’E around 0815.The target seamount had an east-west trending ridge that rose to  a depth of approximately 1100m. The ROV arrived on bottom approximately 100m below the summit. It traversed up a steep slope which was heavily sedimented and had relatively low biomass.  We observed sparse assemblages of hexactinellid sponges, corals with brittle stars and various types of crabs, urchins, and fish. Shallower depths closer to the summit were also heavily sedimented and the feature overall resembled some of the other deeper seamounts we had explored on previous dives.

July 13:
The crew launched the ROV for Dive 13 at 2º 50.651'N 125º 03.503’E around 0500. Initial target depth was 432m.  This was our second dive at a shallow volcanic cone that rises to a depth of 500m. The ROV reached bottom on the south slope at a depth of 650m and moved upslope toward the area where we previously observed a wide variety of corals, crinoids, sponges, crabs, gastropods and fish. We then descended deeper to a depth of 800m and followed a ridge from the southwest back upslope. A much lower biomass and diversity of organisms, consisting mostly of corals with brittle stars and urchins, were visible in the deeper area. There appeared to be a significant amount iron staining around cracks in many of the rocks and what might have been some yellow bacterial mat. As we ascended back upslope biomass began to increase until we were again in an area of higher biological diversity.  NOTE: Several NOAA Leadership representatives observed the ROV dive from the Silver Spring ECC. A more formal event in the Silver Spring ECC will occur on July 23 local shiptime / July 22 Eastern.
SURVEY/CTD:
EM 302 multibeam mapping continued to complement ROV dives. This bathymetry is critical to ROV and CTD target selection. A significant amount of mapping was completed in the area to the south of the western priority box. Some additional mapping in this area remains for Leg III. The Chief ET implemented a short-term fix for the Expendable Bathymetry Thermograph (XBT) issue described in the previous SITREP.
SHIP NETWORK

OMAO shore-side support personnel continue to consider low-speed network issues. No recent progress has been made 
High-speed Network –Since June 20, OER has been paying $6K per day to increase bandwidth from 10Mbps to 20 Mbps to increase the size and quality of the video feeds to the ECCs.  Up until this point the video team and Ship’s ETs have had to reconfigure the network to allow video to be broadcasted at a higher quality.  The ETs have confirmed with MTN that the link to the Okeanos Explorer is 20Mbps and the OMAO network engineer has confirmed the system should support outbound traffic up to 20 Mbps.  Though OMAO is responsible for solving the problem, OER is responsible for paying all MTN contract costs. By July 6th, OER had spent $96K for services that it is unable to use because of the unknown problem.  On that day, the problem apparently disappeared. The team does not consider the problem resolved and continues to work with MTN and NOAA IT to find the root of the problem. 

The ship has started a dialog with NOAANet on connectivity improvements over the high speed link.  These improvements are to enable addition services relate to conducting remote exploration such as remote server management and real-time text-based collaboration. The next step in the process is to gain change approval from OAR CIO and OMAO fleet network security.  OMAO resumed a dialog with the ship and OER personnel to implement changes to network traffic policies.  Progress remains slow, encumbered by personnel availability due to time-zone differences.  There has been no recent progress toward contracting network engineer to examine and trouble-shoot the network from shore. 

Telepresence – Telepresence connectivity began slowly degrading around July 10. Symptoms included frequent audio and video dropouts as well as lower quality video. Unfortunately the symptoms were more visible to shore-side personnel initially. Consequently, shipboard personnel did not begin troubleshooting in the situation worsened. Shipboard troubleshooting began in earnest on the evening of July 12. After the Chief ET spoke with MTN, they determined the issue was originating from Steel Valley and not the Okeanos Explorer.  Though connectivity problems continued on July 13, they were significantly improved over July 12. 

Dynamic Positioning

DP functioned well throughout all of Leg II. There are no issues to report.
============================================================================================================================================================
PLAN OF THE DAY

Thursday, July 15, 2010
In-Port in Bitung
==============================================================================
=========================END OF SITUATION REPORT==============================
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Stalked Sponge from the July 5 dive





Crinoid on sponge from the July 6 dive











�What’s the latest with this?





